REVIEW ARTICLE



Current Agents in Development for Treating Behavioral and Psychological Symptoms Associated with Dementia

Mehnaz Ahmed^{1,2} · Marlene Malik¹ · Johannes Teselink¹ · Krista L. Lanctôt^{1,2,3,4} · Nathan Herrmann^{1,3,4}

Published online: 8 April 2019 © Springer Nature Switzerland AG 2019

Abstract

Behavioral and psychological symptoms associated with dementia are highly prevalent and are associated with an increased risk of institutionalization and mortality. Current pharmacological treatments for these symptoms include cholinesterase inhibitors, antipsychotics, and selective serotonin reuptake inhibitors. When used for treating behavioral and psychological symptoms associated with dementia, they are associated with limited efficacy and/or serious adverse events. As such, there has been extensive research into novel agents with varying mechanisms of action targeting behavioral and psychological symptoms associated with dementia. In this article, we present the results of a comprehensive literature search and review that evaluates current agents that have completed or are currently in clinical trials for treating behavioral and psychological symptoms associated with dementia as a primary outcome. We highlight novel agents from miscellaneous drug classes, such as dextromethorphan/quinidine, bupropion/dextromethorphan, lumateperone, deudextromethorphan/quinidine, methylphenidate and scyllo-inositol, and drugs from various therapeutic classes (including atypical antipsychotics, selective serotonin reuptake inhibitors, and cannabinoids) that have demonstrated promising results and were generally well tolerated. Future research with large appropriately powered studies using validated outcome measures for behavioral and psychological symptoms associated with dementia should be conducted to further establish the clinical utility of these agents.

Key Points

Discovering new medications for treating behavioral and psychological symptoms associated with dementia is an active area of research.

We discuss results from recently completed clinical trials (e.g., methylphenidate, nabilone, dronabinol, and citalopram) and review medications that are still being investigated in clinical trials (such as dextromethorphan/quinidine, bupropion/dextromethorphan, lumateperone, deudextromethorphan, scyllo-inositol, brexpiprazole, and pimavanserin).

Results from recent advances can inform future research into this field and potentially improve treatment.

1 Introduction

There are close to 50 million people worldwide living with some form of dementia [1]. Major forms of dementia include Alzheimer's disease (AD), vascular dementia, dementia with Lewy bodies, and frontotemporal dementias [1]. In its early stages, dementia consists of significant deterioration in cognitive functions and capabilities. These impairments include deficits in memory, such as working, episodic, and semantic memory, processing information, executive functioning, judgment, learning capacity, and language, all of which decline as the disease progresses [2, 3]. Alzheimer's disease is the most common form affecting roughly 60–70% of people with dementia and it is anticipated to affect 115 million people by 2050 [3].

Nathan Herrmann nathan.herrmann@sunnybrook.ca

¹ Neuropsychopharmacology Research Group, Hurvitz Brain Sciences Program, Sunnybrook Research Institute, Sunnybrook Health Sciences Centre, Toronto, ON, Canada

² Department of Pharmacology and Toxicology, University of Toronto, Toronto, ON, Canada

³ Department of Psychiatry, University of Toronto, Toronto, ON, Canada

⁴ Geriatric Psychiatry, Sunnybrook Health Sciences Centre, Toronto, ON, Canada

In addition to cognitive impairment, individuals affected with dementia often experience neuropsychiatric symptoms, also known as behavioral and psychological symptoms related to dementia (BPSD) [2, 3]. These symptoms include depression, irritability, hallucinations, delusions, sleep disturbances, eating problems, anxiety, apathy, agitation, and aggression [4]. Moreover, clusters of BPSD, such as hyperactivity (agitation and aggression), affective disorder (anxiety and depression), and psychosis (delusions and hallucinations) are common [5]. A study by Steinberg et al. estimated the 5-year point prevalence of BPSD in a sample of 408 patients with dementia, with symptoms assessed annually starting at baseline [6]. They found that the point prevalence for delusions was 18% at baseline compared to 34–38% at year 5 [6]. Additionally, the point prevalence of hallucinations went from 10 to 19-24%, agitation from 13 to 24%, depression from 29% at baseline compared to 41-47%, apathy from 20 to 51%, and anxiety from 14 to 24–32% [6]. The point prevalence for any symptom was 56% at baseline, and 76-87% at year 5. The 5-year point prevalence was greatest for depression at 77%, followed by apathy (71%) and anxiety (62%) [6].

Over the course of the disease, 98% of those with dementia experience at least one neuropsychiatric symptom [3, 6, 7]. This not only reduces the quality of life for the patients, but also places significant distress on the caregiver and the patients' family [8]. Economically, the impact of BPSD on healthcare costs is significant, with severity of symptoms being associated with increases in the cost of care [9, 10]. The presence of BPSD has been shown to be a significant predictor of a patient becoming institutionalized, requiring extensive care, and having an increased risk of mortality [4, 5].

The diagnostic criteria for dementia produced by the National Institute of Aging, American Psychiatric Association's Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition and the National Institute of Neurological and Communicative Disorders and Stroke and Alzheimer's Disease and Related Disorders Association have been pivotal tools in improving treatments in clinical research settings based on significant improvements in understanding the etiology and neuropathological changes associated with dementia. Furthermore, advocacy for biomarker-based diagnostic criteria for AD has stimulated further research into novel and personalized treatment protocols.

The prevalence and impact of BPSD have led to recognition of the need for pharmacological interventions. The pharmacological management of BPSD requires an understanding of the medical, social, and psychological context in which the symptoms present, and usually occurs after the implementation of non-pharmacological treatments as firstline interventions [11]. Professional healthcare providers must first control for precipitating factors that may influence BPSD prevalence, such as environmental triggers, comorbidities, and pain [5]. When non-pharmacological interventions prove to be difficult to implement or ineffective in reducing symptoms, pharmacological treatments can be considered. Many current pharmacological agents have been repurposed in clinical trials for treating BPSD. The traditional agents aimed at treating cognitive symptoms of dementia are cholinesterase inhibitors and memantine. Cholinesterase inhibitors, which have modest benefits for cognition, can improve neuropsychiatric symptoms [12]. Atypical antipsychotics, such as risperidone, olanzapine, and aripiprazole, have demonstrated efficacy for improving agitation, aggression, and psychosis symptoms associated with dementia [13–15]. Symptoms of depression and anxiety are often treated with selective serotonin reuptake inhibitors (SSRIs) [16]. Similarly, benzodiazepines have been prescribed for the management of anxiety on an acute basis [16].

While BPSD can sometimes be managed with pharmacotherapies, these agents are associated with adverse events, requiring careful consideration of the risk/benefit ratio. The Clinical Antipsychotic Trials of Intervention Effectiveness-Alzheimer's Disease trial compared olanzapine, quetiapine, risperidone, and placebo, finding that those atypical antipsychotics had similar efficacy for improving BPSD but resulted in cognitive decline over time [15]. Numerous meta-analyses have concluded that atypical antipsychotics can increase the risk of cerebrovascular accidents and mortality in patients with dementia and these agents therefore have black-box warnings in this population [17]. Some SSRIs also carry a risk of experiencing cardiovascular events; the Citalopram for Agitation in Alzheimer Disease (CitAD) study found that a 30-mg/day dose of citalopram resulted in prolonged QT intervals [18–20]. The adverse events associated with benzodiazepines include sedation, ataxia, falls, and negative cognitive effects; hence there is wariness with prescribing these agents long term in elderly populations [16, 21].

The treatment of BPSD is dependent upon accurately and specifically measuring these symptoms. There are numerous BPSD measures that have been validated for symptoms like depression, agitation, and psychosis such as the Neurobehavioral Rating Scale (NBRS) and the Neuropsychiatric Inventory (NPI) [22]. The NPI is most notably used to measure the 12 BPSD associated with AD and dementia and has a version available to be used in nursing homes known as the Neuropsychiatric Inventory-Nursing Home (NPI-NH). The NPI furthermore includes caregiver distress as a measure to look at the secondary effects of behavior on caregivers. Scales for depression include the Geriatric Depression Scale, the Center for Epidemiologic Studies Depression Scale, the Hamilton Depression Rating Scale, and the Cornell Scale for Depression in Dementia [23, 24]. Behavioral disturbances such as agitation and aggression can be measured by the Behavioral Pathology in Alzheimer's Disease [25], Cohen Mansfield Agitation Inventory (CMAI) [26], NPI-Agitation/ Aggression Subscale [27], and Pittsburgh Agitation Scale [28]. Apathy is often measured by the Apathy Evaluation Scale (AES) as well as with the Apathy Subscale of the NPI [29].

The NPI is often used in clinical trials because of its comprehensiveness, lack of symptom overlap, ease of administration, and flexible approach to observing BPSD [22]. Studies utilizing the NPI have examined the rate of behavioral change annually, relationship to cognitive, global, and functional measures, standard deviations, and also requirements for determining sample size in clinical trials [22]. As a result, its use in clinical trials has good content validity with internal consistency, with positive test–retest and inter-rater reliability across various groups [4, 22].

Given that there are few options for treating BPSD, further research into understanding the etiology of these symptoms in association with AD and dementia, and the allocation of resources towards developing novel agents that uniquely target neuropsychiatric symptoms is warranted. Fortunately, a new era of pharmacological interventions targeting novel methods of treating BPSD may shed light on providing care for patients with dementia, while reducing the risk of adverse events. Therefore, this review intends to focus on current and novel agents being investigated for treating BPSD as the primary outcome in clinical trials.

Three reviewers individually searched ClinicalTrials. gov for investigational agents in completed and ongoing clinical trials explored for the treatment of neuropsychiatric symptoms associated with dementia (up to July 2018). We searched the terms "depression", "irritability", "hallucinations", "delusions", "anxiety", "apathy", "agitation", "aggression", "sleep disturbances", and "eating disorders" in conjunction with "neuropsychiatric symptoms" or "behavioral and psychological symptoms related to dementia" for the condition "dementia". We included studies where BPSD were evaluated as a primary outcome, using validated rating scales for the symptom in question. We included agents that were explored in randomized, double-blind, placebo-controlled trials. We excluded devices that were being explored for BPSD and phase IV studies. We then completed a comprehensive search of the literature using the OVID databases for published clinical trial results and associated studies of the included agents and inquired into findings reported by the pharmaceutical company websites when published results were unavailable.

The advantages of this approach include the assurance that all studies included in this review were randomized, double-blind, placebo-controlled studies and allowed us to discuss ongoing studies investigating agents for which results have not yet been published. A disadvantage of this approach is that we could not conduct this review as a systematic review or meta-analysis. Table 1 summarizes study characteristics of ongoing clinical trials included in this review.

2 Atypical Antipsychotics

Up to 60% of patients with dementia in hospitals and longterm care homes are prescribed antipsychotics to treat neuropsychiatric symptoms related to dementia and AD [5, 14, 30]. The majority of these antipsychotics are second-generation atypical antipsychotics; while having been studied extensively, further research into these agents is necessary given the effect sizes in previous studies performed, poor drug tolerability, and increased risk of mortality related to their use [5, 30]. Newer agents that have been explored in this class include brexpiprazole, lumateperone, and pimavanserin.

Brexpiprazole was originally approved by the US Food and Drug Administration in July 2015 to treat adults with schizophrenia and as an added treatment with antidepressants to treat adults with major depressive disorder. It is a quinolinone derivative of aripiprazole and its mechanism of action is not fully elucidated. Its efficacy is hypothesized to be mediated by a combination of partial agonism at dopamine D₂ receptors and at serotonin 5-HT_{1A} receptors, as well as antagonist activity at 5-HT_{2A} receptors with high affinity, with additional antagonistic affinity for α 1B/2C adrenergic receptors [31, 32].

Two phase III, randomized, double-blind, placebo-controlled, multicenter studies with 700 individuals in total, examined the efficacy, safety, and tolerability of brexpiprazole. Both studies recruited outpatients and institutionalized patients aged 51-90 years diagnosed with probable AD and symptoms of agitation, who were required to have a total score of ≥ 4 on the agitation/aggression item of the NPI at screening and baseline to be included in the study. Participants were assigned fixed doses of 1 mg/day, 2 mg/ day, or placebo in one trial, and 0.5 mg/day, 1 mg/day, or 2 mg/day or placebo in the other trial [33]. Primary outcome measures focused on changes from baseline in the CMAI total score after 12 weeks. In the first trial (n = 430), patients demonstrated significantly reduced CMAI scores compared with placebo [33]. However, while the second study (n = 270) showed an overall reduction in total CMAI in the brexpiprazole group, it was not significantly different compared to placebo [33]. Adverse events in both trials were reported with an incidence of 3% or greater in active vs. control groups, with insomnia (4.7% vs. 3.3%, respectively), agitation (3.5% vs. 2.9%), and somnolence (3.3% vs. 2.2%) being the most common adverse events [33]. There

Table 1 Agents (Table 1 Agents currently being investigated in ongoing clinical trials for behavioral and psychological symptoms associated with dementia	tigated in	n ongoing clinical tr.	ials for behavid	oral and psychologic	al symptoms associ	iated with demen	itia		
Agent	Clinical trial identifier	Phase	Participant diag- nosis	Number of participants	Study design	Dose	Study duration	Behavioural or psychiatric symp- tom targeted	Primary outcome	Tentative study completion date
Brexpiprazole	NCT03620981	H	Clinical diagnosis of AD	407	Randomized, double-blind, placebo-con- trolled, parallel- group	Low dose or high dose once daily	10 weeks	Agitation	Change in CMAI score from baseline to 10 weeks	December 2020
Lumateperone	NCT02817906	Ξ	Clinical diagnosis of AD	360	Randomized, double-blind, placebo-con- trolled, parallel- group	9 mg/day	4 weeks	Agitation	Change in CMAI score over 4 weeks	August 2018
Escitalopram	NCT03108846	Ξ	Clinical diagnosis of AD, based on NIA-AA criteria	392	Randomized, double-blind, placebo-con- trolled, parallel- group	5–15 mg/day	12 weeks	Agitation	Modified Alzhei- mer's Disease Cooperative Study-Clinical Global Impres- sion of Change over 12 weeks	August 2022
Cannabis oil	NCT03328676	н	Diagnosis of dementia (NCD), based on DSM-V cri- teria for at least 6 mo prior	60	Randomized, double-blind, placebo-con- trolled, parallel- group	Each Avidekel oil drop is approxi- mately 0.04 mL in volume con- taining about 12 mg of CBD and 0.6 mg of 9-THC	16 weeks	Agitation	Proportion of subjects achieving a CMAI \geq 4-point decrease during the treatment period	May 2020
Dronabinol	NCT02792257	=	Diagnosis of dementia due to AD	160	Randomized, double-blind, placebo-con- trolled, parallel- group	5–10 mg/day	3 weeks	Agitation	Symptoms of agitation over a 3-week time frame	December 2020

Agent	Clinical trial	Phase	Phase Participant diag-	Number of	Study design	Dose	Study duration	Behavioural or	Primary outcome	Tentative study
			SIGUI	participatits				psycination symp- tom targeted		comprehendin date
AVP-786	NCT03393520	Ξ	Diagnosis of probable AD based on NIA- AA criteria	412	Randomized, double-blind, placebo-con- trolled, parallel- group	2 unspecified daily AVP-786 doses	12 weeks	Agitation	Change in CMAI from baseline to week 12	June 2021
	NCT02442778	∃	Diagnosis of probable AD based on NIA- AA criteria	470	Randomized, double-blind, placebo-con- trolled, parallel- group	2 unspecified daily AVP-786 doses	12 weeks	Agitation	Change in CMAI from baseline to week 12	December 2019
	NCT02442765	∃	Diagnosis of probable AD based on NIA- AA criteria	380	Randomized, double-blind, placebo-con- trolled, parallel- group	2 unspecified daily AVP-786 doses	12 weeks	Agitation	Change in CMAI from baseline to week 12	April 2019
Methylphenidate NCT02346201	NCT02346201	⊟	Possible or prob- able AD, based on NINCDS- ADRDA criteria	200	Randomized, double-blind, placebo-con- trolled, parallel- group	20 mg/day	6 months	Apathy	Change in NPI-Apathy Subscale score from baseline to 6 months	August 2020
Lithium	NCT02862210	=	Behavioral variant FTD or semantic variant, primary progressive aphasia, which is generally accompanied by a behavioral syndrome, or agrammatic/ non-fluent pri- mary progres- sive aphasia with behavioral symptoms	60	Randomized, double-blind, placebo-con- trolled, parallel- group	600 mg/day	12 weeks	Agitation and aggression	Change in NPI score over 12 weeks	March 2020
	NCT02129348	Ξ	Possible or prob- able AD by standard NIA criteria	80	Randomized, double-blind, placebo-con- trolled, parallel- group	600 mg/day	12 weeks	Agitation	Change in NPI score over 12 weeks	April 2019

593

Table 1 (continued)	led)									
Agent	Clinical trial identifier	Phase	Phase Participant diag- nosis	Number of participants	Study design	Dose	Study duration	Study duration Behavioural or psychiatric symp- tom targeted	Primary outcome	Tentative study completion date
Carbamazepine Mirtazapine	NCT03031184	E	Possible or prob- able AD based on NINCDS- ADRDA criteria	471	Randomized, double-blind, placebo-con- trolled, parallel- group	100 mg/day 15 mg/day	12 weeks	Agitation	Change in CMAI score from baseline to 12 weeks	August 2019
MP-101	NCT03044249	=	Dementia associated with Parkinson's disease, demen- tia with Lewy bodies, possible or probable AD, frontotemporal degeneration spectrum disor- ders, vascular dementia	100	Randomized, double-blind, placebo-con- trolled, parallel- group	60 mg/day	10 weeks	Psychosis and/ or agitation and aggression	Change from baseline to week 10 in NPI-Psychosis Subscale	January 2021
AXS-05	NCT03226522	111/11	Diagnosis of probable AD, based on NIA- AA criteria	435	Randomized, double-blind, placebo-con- trolled, 3-arm, parallel-group	Unspecified	5 weeks	Agitation	Change from baseline to week 10 in the CMAI	September 2019
Intranasal oxytocin (Syntocinon [®])	NCT03260920	П	Diagnosis of probable FTD (behavioural variant FTD, FTD-semantic subtype, or FTD-progres- sive non-fluent aphasia)	112	Randomized, double-blind, controlled, trve tive	Unspecified low, medium, and high dose	20 weeks	Apathy	Change in NPI apathy score over 20 weeks	September 2021
AD Alzheimer's dementia, NCD I and Communicat	disease, <i>CBD</i> canr reurocognitive disor ive Disorders and S	abidiol, e rder, <i>NIA</i> troke-Alz	AD Alzheimer's disease, CBD cannabidiol, CMAI Cohen Mansfield Agitation Inventory, DSM-V Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition, FTD frontotemporal dementia, NCD neurocognitive disorder, NIA National Institute on Aging, NIA-AA National Institute on Aging and Alzheimer's Association NINCDS-ADRDA National Institute of Neurological and Communicative Disorders and Stroke-Alzheimer's Disease and Related Disorders Association, NPI Neuropsychiatric Inventory, THC tetrahydrocannabinol	eld Agitation Aging, <i>NIA</i> A Related Disv	1 Inventory, DSM-V AA National Institute orders Association, 1	Diagnostic and St e on Aging and Alz <i>NPI</i> Neuropsychiati	atistical Manual theimer's Associ ric Inventory, TH	of Mental Disorder ation <i>NINCDS-ADR</i> , 'C tetrahydrocannabi	s, Fifth Edition, <i>FT</i> <i>DA</i> National Institut inol	D frontotemporal e of Neurological

were 66 deaths occurring during the trial; however, those were judged to be unrelated to treatment [33]. A phase III randomized, double-blind, placebo-controlled trial is currently underway (n=407) where participants with probable AD will receive 1 or 2 mg/day of brexpiprazole or placebo over 10 weeks to treat agitation with the primary outcome measure being mean change in CMAI scores from baseline at week 10 (ClinicalTrials.gov Identifier: NCT03620981).

Lumateperone is a newer investigational atypical antipsychotic. It is a potent antagonist at 5-HT_{2A} receptors, and a serotonin reuptake inhibitor. It further exhibits presynaptic agonism and postsynaptic antagonism at D₂ receptors, which allows it to act as a mesolimbic/mesocortical dopamine phosphoprotein modulator [34–36]. Furthermore, it binds with high affinity to D₁ receptors [36, 37]. It indirectly enhances glutamatergic modulation through N-methyl-D-aspartate and α -amino-3-hydroxy-5-methyl-4isoxazolepropionic acid channels in the prefrontal dorsolateral cortex by D₁ receptor activation and increases protein phosphorylation in the mammalian target of rapamycin pathway [35, 38].

Its ability for indirect glutamatergic modulation, and multiple receptor pathway activity, suggests its potential use as a rapid treatment agent in treating a broad range of BPSD such as agitation, aggression, and sleep disturbances with a reduced incidence of side effects compared with other atypical antipsychotics. Initial phase II trial results showed it was well tolerated across a range of low dosages with improved cognition and reductions in sleep disturbances, agitation, and depression [37, 39]. A phase III, double-blind, placebocontrolled, multicenter trial is currently underway looking at the efficacy and safety of a 9-mg/day dosage in patients with AD (n = 360) for 4 weeks (ClinicalTrials.gov identifier: NCT02817906). Patients recruited will have a clinical diagnosis of probable AD and clinically significant symptoms of agitation, with the primary outcome measure being the change in CMAI score.

Pimavanserin was approved in 2016 for treating hallucinations and delusions with psychosis in Parkinson's disease, showing greater efficacy in patients with more severe psychotic symptoms [40]. Interestingly, it works as an inverse agonist and antagonist at 5-HT_{2A} receptors, reducing the baseline activity of the 5-HT_{2A} receptor and blocking the actions of agonists. More recently, pimavanserin has been shown to mediate effects at 5-HT_{2C} receptors without any antagonistic properties at D₂ receptors—in contrast to many antipsychotics in treating psychosis [40, 41]. Because an upregulation of 5-HT_{2A} receptors is implicated in psychotic symptoms, pimavanserin working as an inverse agonist at 5-HT_{2A} receptors may lead to a preservation in cognition and a reduction in psychotic symptoms, with the minimized 5-HT_{2C} activity mitigating side effects associated with atypical use [40, 42].

Pimavanserin has recently been examined as a treatment for neuropsychiatric symptoms in AD. A phase II, randomized, double-blind, placebo-controlled trial across the UK included patients (n = 181; 90 receiving pimavanserin) with possible or probable AD and psychotic symptoms such as delusions or hallucinations, who received either 34 mg/day of pimavanserin or placebo. Primary outcome measures focused on change from baseline at week 6 measured by the NPI-NH, as well as maintained efficacy and safety at week 12. At week 6, the pimavanserin group had a significantly greater mean change (39.5%) in their NPI-NH psychosis score from baseline compared with the control group (19.3%) [41]. Fifty-five percent of pimavanserin-treated individuals saw a clinically relevant response of greater than 30% improvement in psychosis subscale scores and this was significantly different compared to 37% of controls [41]. Results in other individual domains of NPI were not significant. The data were also stratified by comparing patients who scored either < 6 and ≥ 6 on the Mini-Mental State Examination and <12 and >12 on the NPI-Psychosis Subscale at baseline. The high-psychosis group saw significantly more improvement in symptoms as demonstrated by a greater mean change in the NPI-NH psychosis score in the pimavanserin vs. placebo group [41]. Pimavanserin did not significantly reduce NPI-NH psychosis scores compared to baseline scores in patients with less severe psychotic symptoms at baseline (<12 on the NPI-Psychosis Subscale) compared to placebo [41]. The NPI total score was not significantly different between treatment and control groups. Unfortunately, there was no observable difference between treatment and control groups at week 12, which could represent a lack of sustained benefit, but could also be attributable to the heterogeneous and dynamic nature of psychosis over time [30, 41].

Common side effects in the active vs. control groups included falls (23% and 23%), urinary tract infections (22% and 28%), and agitation (21% and 14%), with eight participants in the active group and 11 participants in the control groups discontinuing because of adverse events [41]. Currently, a double-blind, placebo-controlled, relapse-prevention study of pimavanserin (n = 356) is underway to evaluate the efficacy of pimavanserin in preventing a relapse of psychotic symptoms in subjects with dementia-related psychosis who were stabilized after 12 weeks of an open-label pimavanserin treatment of either 20 mg, 34 mg, or placebo.

3 Selective Serotonin Reuptake Inhibitors

Substantial evidence shows that monoaminergic systems are affected in the pathogenesis of AD [43], and this has been proposed as a target for BPSD [44–47]. In particular, the serotonergic system [44, 48, 49] has been targeted in recent clinical trials using selective serotonin reuptake inhibitors (SSRIs) as an alternative to antipsychotics to treat

neuropsychiatric symptoms such as psychosis, agitation, aggression, and depression. Selective serotonin reuptake inhibitors are generally well tolerated in patients with AD, resulting in fewer safety concerns and decreased risks to mortality compared with antipsychotics and other antidepressant medications [50].

Citalopram is one of the most selective inhibitors of the SSRI class and demonstrates a low incidence of side effects compared with other SSRIs, good patient compliance, and a low potential for interactions with other concomitant medications [51]. Based on these factors, citalopram has been compared to the antipsychotics perphenazine, risperidone, quetiapine, and olanzapine for the treatment of BPSD, particularly agitation and aggression. Patients who received citalopram or perphenazine showed statistically significant improvements on several Neurobehavioral Rating Scale factor scores compared with placebo, though only patients treated with citalopram showed significantly increased improvement in both their total Neurobehavioral Rating Scale and agitation/aggression scores compared with the placebo group [52]. In the study by Pollock et al., agitated/ aggressive and psychotic symptoms decreased in both the citalopram and risperidone groups, though the improvement between the two groups did not differ significantly and the study lacked a placebo group. Notably, there was a significant increase in side-effect burden with risperidone but not with citalopram [53].

Similarly, Viscogliosi et al. found similar efficacy between citalopram, quetiapine, and olanzapine in reducing agitation with less adverse outcomes in the citalopram group compared with the atypical antipsychotics [54]. Based on these studies, the evidence supporting the efficacy of SSRIs in treating BPSD was intriguing but limited and required continued investigation.

The CitAD study was conducted to further explore the efficacy of citalopram for treating agitation in non-depressed patients with dementia [19]. The CitAD study was a phase III, randomized, double-blind, placebo-controlled, multicenter, parallel-group clinical trial in which participants with probable AD and clinically significant agitation received standardized psychosocial interventions combined with either citalopram up to 30 mg daily (n = 94) or placebo (n=92). The citalopram group showed a significant reduction in agitation compared with placebo after 9 weeks of treatment; specifically, the citalopram group had significant improvements on the NBRS Agitation Subscale, CMAI, and total NPI [19]. Overall, there was no difference between the citalopram and placebo groups in terms of adherence or serious adverse events, though the citalopram group did show a significantly increased frequency of adverse events such as anorexia and fever. The citalopram group demonstrated a significantly greater decline in cognition as measured by lower Mini-Mental State Examination scores and an increase in the corrected QT interval by electrocardiogram [19]. Because of concerns regarding prolonged corrected QT intervals in the elderly caused by citalopram, the US Food and Drug Administration recommended dose restrictions while the CitAD study was ongoing [19]. As the CitAD study and previous studies utilized higher doses of citalopram, it was deemed difficult to determine if lower doses of citalopram would be efficacious in clinical practice [50]. Nonetheless, patients and their caregivers may find citalopram to be a more desirable pharmacological intervention compared to antipsychotics; the potential decline in cognition and increase in corrected QT intervals associated with citalopram are less severe than the heightened risk of cerebrovascular problems and mortality linked to antipsychotics [55].

Citalopram enantiomers have different affinities for the serotonin transporter; (S)-citalopram has about a 40-fold greater affinity compared with (R)-citalopram [56]. Microdialysis data have demonstrated that (S)-citalopram but not (R)-citalopram increases extracellular 5-HT levels in the rat frontal cortex [57]. Because (R)-citalopram blocks (S)-citalopram from binding at the transporter in vitro, it is hypothesized that (R)-citalopram may reduce the therapeutic benefit of citalopram if in fact the SSRI treats agitation by targeting the serotonergic system [55]. Through populationpharmacodynamic models based on data from the CitAD study, (R)-citalopram is speculated to have accounted for a greater amount of the adverse effects associated with citalopram while (S)-citalopram (escitalopram) exposure was more efficacious for the treatment of agitation [56]. Wu et al. found that when treated for major depressive disorder, geriatric patients who received escitalopram experienced superior treatment persistence and less hospitalizations compared with patients who received citalopram [57].

Escitalopram was compared to the atypical antipsychotic risperidone for the treatment of BPSD associated with AD in a pilot, 6-week randomized, double-blind, controlled trial for in-patients hospitalized owing to behavioral symptoms of AD (n=40) [58]. Patients received either 1 mg of risperidone or 10 mg of escitalopram per day. The NPI total score improved in both groups, successfully reducing psychotic symptoms and agitation with no significant differences between groups. Completion rates were higher for escitalopram (75%) compared with risperidone (55%) as a result of adverse events, of which there were none reported for the escitalopram group [58]. A 12-week, randomized, doubleblind trial for patients with depression associated with AD (final n = 60) found no significant differences in measures of depression between the escitalopram and placebo groups as indicated by scores on the Cornell Scale for Depression in Dementia [53]. Adverse events related to treatment did not significantly differ between the two groups and no serious adverse events related to treatment were reported. Despite the promising results found by An et al., 29% of participants enrolled did not finish the study; future studies will require increased statistical power before meaningful conclusions can be drawn [59]. Additionally, An et al. discussed the need to recruit patients whose depressive symptoms are characterized more robustly or as more severe to improve statistical power [59].

Findings regarding the efficacy of escitalopram indicate a potential alternative with less adverse effects compared to antipsychotics in treating agitation, psychotic symptoms, and depression. These findings must be replicated in future studies for longer durations and in larger samples before escitalopram can be recommended as an alternative to antipsychotics in clinical practice [50]. Currently, a phase III, 12-week, randomized, double-blind study using escitalopram for the treatment of clinically significant agitation in patients with AD is underway with a projected sample size of 392 patients (ClinicalTrials.gov identifier: NCT03108846).

4 Cannabinoids

It has recently been postulated that cannabinoids may be useful for treating agitation/aggression symptoms in dementia populations, based on case reports and retrospective and open-label clinical studies [60, 61]. They act on CB₁ and CB⁻₂ cannabinoid receptors in the central nervous system mediating appetite, sedation, motor activity and coordination, and anti-anxiety effects [62–64]. However, previous clinical trial results have been mixed, warranting the need for larger, more robust studies to warrant the use of these agents for treating neuropsychiatric symptoms.

Delta-9-tetrahydrocannabinol (THC) itself has been explored for treating agitation in patients with AD. In a placebo-controlled, parallel-assignment trial where 24 participants were taking either a low dose of THC (1.5 mg) or a high dose of THC (4.5 mg) daily for 3 weeks, there was no difference in NPI or CMAI scores between the treatment groups [65]. Currently, there is an ongoing 16-week, randomized, double-blind, placebo-controlled trial investigating the efficacy of a different formulation of THC in the form of cannabis oil, containing 12 mg of cannabidiol and 0.6 mg of delta-9-THC per drop taken three times daily, in reducing agitation/aggression in patients defined as having dementia based on the Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition (ClinicalTrials.gov identifier: NCT03328676). The trial intends to recruit 60 patients with dementia with an NPI-NH agitation/aggression score of \geq 3, and the primary endpoint is the proportion of subjects achieving a decreased score of 4 or more on the CMAI at the end of 16 weeks. Prior to this study, an open-label 4-week trial found that medical cannabis oil at a 2.5-mg dose taken twice daily significantly reduced delusions, agitation/aggression, irritability, apathy, and sleep and caregiver distress on the NPI in patients with AD [66].

Dronabinol, a synthetic THC analog and full CB₁ and CB⁻₂ agonist, was the first cannabinoid explored in a randomized, placebo-controlled, crossover trial for this indication. A twice-daily dose of 2.5 mg was found to be efficacious in reducing disturbed behavior measured by the CMAI in patients with AD who were refusing food [67]. A similar clinical trial with 2.5 mg of dronabinol resulted in decreased night-time activity, as measured by wrist actigraphy, in the two patients enrolled in the trial [68]. Additionally, NPI scores decreased in both patients [68]. In both trials, there were carry-over effects of dronabinol as there were no washout periods. In the first dronabinol trial by Volicer et al [67] adverse events such as euphoria and fatigue were more frequently reported during dronabinol treatment; whereas the two patients in the latter trial reported no adverse effects [67, 68]. The dronabinol trials had promising results but are limited by their small sizes and carry-over effects. Forester et al. are currently recruiting for a larger double-blind, placebocontrolled, parallel-group trial to explore the efficacy of this agent further in agitated inpatients with AD (ClinicalTrials. gov identifier: NCT02792257). They are employing a larger dose than used in previous trials (10 mg daily) as it may be more effective in reducing agitation, which will be measured using the Pittsburgh Agitation Scale and NPI-Clinician Rating Scale [69].

Recently, nabilone, a synthetic cannabinoid and partial CB1 and CB22 agonist, significantly improved CMAI and NPI-agitation/aggression scores in agitated patients with AD in a randomized, placebo-controlled, crossover trial with 39 participants [70]. Overall NPI scores significantly improved during nabilone treatment and the Clinical Global Impression Scale indicated that patients had greater improvement while taking nabilone [70]. Nabilone was well tolerated in these patients. Sedation was a more common adverse effect while taking nabilone but was not significantly different between the treatment groups [70]. Presently, based on clinical trial results, nabilone appears to be a promising therapeutic cannabinoid for treating agitation and aggressive symptoms due to AD, given its advancement in the drug development pipeline. The results of the nabilone trial warrant the implementation of a larger multi-center study to further validate these results.

5 Dextromethorphan/Quinidine Agents

Dextromethorphan/quinidine, formulated in combination with low-dose quinidine, was examined to treat agitation and aggression in patients with probable AD. Normally marketed as an antitussive, dextromethorphan is a low-affinity N-methyl-D-aspartate receptor antagonist, $\sigma 1$ receptor agonist, serotonin and norepinephrine reuptake inhibitor, and neuronal nicotinic $\alpha 3\beta 4$ receptor antagonist [71]. The ability for dextromethorphan to exert its effects in the central nervous system is limited because it is metabolized by the liver enzyme cytochrome P450 2D6; hence, the rationale for formulating it with quinidine. Quinidine, an anti-arrhythmic agent that blocks voltage-gated sodium channels, inhibits cytochrome P450 2D6; thus, increasing the bioavailability of dextromethorphan and making its effects longer lasting [72, 73].

Cummings et al. found that dextromethorphan/quinidine improved NPI agitation/aggression scores compared with placebo in a randomized, placebo-controlled, 10-week, sequential parallel-comparison trial [74]. The premise of the trial design was to improve the power of the study by offsetting placebo effects [74]. In stage 1, 93 participants were randomized to receive 20 mg/10 mg of dextromethorphan/ quinidine once a day for the first week followed by twice a day for weeks 2 and 3, while 127 participants received placebo. From week 4 onward, stage 2 of the trial commenced. Patients who received dextromethorphan/quinidine continued their treatment while participants who did not respond to placebo were once again randomized in a 1:1 ratio to receive either 30 mg/10 mg of dextromethorphan/quinidine twice daily or placebo. In both stages, NPI agitation/aggression scores significantly decreased from baseline in the dextromethorphan/quinidine groups compared with the placebo group [74]. Furthermore, when simulating a parallel-group design comparing participants only randomized to dextromethorphan/quinidine (n=93) and those only randomized to placebo (n = 66) across the 10 weeks, dextromethorphan/ quinidine was superior in improving agitation/aggressive symptoms [74]. There was a significantly greater proportion of participants having clinically meaningful results of at least a 30% reduction in their NPI agitation/aggression scores over 10 weeks when only receiving dextromethorphan/quinidine (65.6%), compared with 47% of patients who only received placebo [74]. Additionally, dextromethorphan/ quinidine significantly improved caregiver distress scores on the caregiver strain index, NPI total scores, NPI irritability/ lability domain scores, and Cornell Scale for Depression in Dementia scores. Dextromethorphan/quinidine was generally well tolerated by participants, although there was a greater incidence of falls while taking this treatment. Larger trials investigating dextromethorphan/quinidine should be conducted to corroborate their findings and further establish the safety profile of this agent.

Deudextromorphan is a newer modified version of dextromethorphan/quinidine. It consists of deuterated dextromethorphan formulated with a low dose of quinidine. By modifying dextromethorphan with deuterium atoms, the chemical bonds are stronger; thus, making this agent less susceptible to enzyme cleavage by cytochrome P450 2D6. With lower first-pass liver metabolism, a much lower dose of quinidine is required in the overall formulation, resulting is a lower potential for drug interactions and cardiac effects induced by quinidine [75]. The drug manufacturer has demonstrated that deudextromorphan has a similar pharmacological profile to dextromethorphan/quinidine [76]. The safety, tolerability, and efficacy of deudextromethorphan is currently being investigated in phase II, placebo-controlled, multi-center clinical trials (ClinicalTrials.Gov Identifiers: NCT02442765, NCT02442778, NCT03393520). The TRIAD-1 and TRIAD 2 studies are currently recruiting participants to investigate two different doses of deudextromorphan and are expected to enrol 380 and 325 patients, respectively [77]. The endpoints are changes in CMAI scores and NPI agitation/aggression scores over 12 weeks [77].

6 Miscellaneous Agents

6.1 Scyllo-inositol

Scyllo-inositol is an endogenous inositol stereoisomer that is hypothesized to bind and inhibit A β 42 peptide aggregates and A β fibrillogenesis, thereby potentially preventing A β deposition and A β -induced toxicity in patients with AD [78]. This drug exhibited promising therapeutic implications for improving symptoms associated with AD in preclinical mice models in which exposure to scyllo-inositol resulted in lower brain A β levels and plaque burden, preserved synaptic density, and improvement in existing learning deficits, and subsequently entered clinical trials [79].

In a 78-week, phase II, double-blind trial, 353 patients with mild-to-moderate AD were randomized to scyllo-inositol (250, 1000, or 2000 mg) or placebo twice daily to investigate the safety, efficacy, and biomarker effects of scylloinositol. No significant difference between the scyllo-inositol 250-mg and placebo groups was found on the Neuropsychological Test Battery after 78 weeks. The 1000- and 2000-mg groups were not included in the primary analysis because of greater rates of serious adverse events, including nine deaths, leading to an early discontinuation of the two higher dosage groups [80]. In addition to observing the 250-mg dose to be safe, scyllo-inositol had favorable pharmacokinetic measures with concentrations found to be higher in cerebrospinal fluid and the brain while cerebrospinal fluid Aβ42 levels significantly decreased compared with the placebogroup. The authors concluded that their study was limited by small sample sizes, rendering the results inadequate in supporting or refuting the efficacy of scyllo-inositol.

A subsequent study evaluated three scyllo-inositol doses (250 mg, 1000 mg, and 2000 mg) in comparison to a placebo in 351 patients with neuropsychiatric symptoms in a 78-week, phase II study for individuals with mild-to-moderate AD. Though differences between treatment and control groups were not significant, the treatment group did experience improved neuropsychiatric symptoms that appeared to be partially mediated by brain myo-inositol regulation [81]. There are no current trials in progress for scyllo-inositol owing to a lack of efficacy during trials leading to termination; however, future studies may target earlier stages of AD and be useful in preventing neuropsychiatric symptoms associated with AD altogether [80].

6.2 Methylphenidate

Methylphenidate is a central nervous stimulant that blocks the dopamine transporter and norepinephrine transporter [82]. Dysfunction in the dopaminergic mesolimbic brain reward system is associated with symptoms of apathy in patients with AD, thus increasing dopamine levels at the level of the synapse may facilitate increases in motivation [83]. In a 12-week open-label study with 23 patients, a significant improvement in apathy was observed as measured on the AES [84]. A prospective, randomized, double-blind, placebo-controlled trial conducted in community-dwelling veterans (n=60) with mild AD demonstrated significantly greater improvements in apathy as measured by AES scores in the methylphenidate vs. the placebo group. Notably, patients in the methylphenidate group experienced greater improvement in functional status and depression when compared with the placebo group [85]. In a 6-week double-blind, multicenter trial (ADMET), 60 patients with AD with apathy were randomized into groups receiving 20 mg of methylphenidate per day or placebo. The methylphenidate group demonstrated significant improvements in NPI apathy scores over 6 weeks of treatment with potential clinically significant effects [86]. Apathy Evaluation Scale scores improved in the methylphenidate group compared with the placebo group though the results were not significant. This may be because of greater variance than seen in prior trials, resulting in a lack of power to detect AES changes within the current study design [86]. Additionally, a trend in global cognition improvements and limited adverse events and side effects were observed in the methylphenidate group. Because of the small sample size and a brief period for follow-up, further research on methylphenidate to treat apathy is planned to determine long-term clinical efficacy in heterogeneous populations [86]. While there is currently limited high-quality evidence for methylphenidate in effectively treating apathy, this agent seems promising as a potential pharmacotherapy for apathy in AD [87]. A follow-up, phase III, double-blind multicenter trial is in progress (ADMET 2) to compare the mean difference in NPI apathy scores from baseline to 6 months in 200 patients with AD randomly assigned to a target dose of 20 mg of methylphenidate daily or a matching placebo (ClinicalTrials.gov identifier: NCT02346201) [88].

6.3 Prazosin

In AD, a significant reduction in norepinephrine-producing locus coeruleus neurons has been observed, resulting in noradrenergic deficits in the hippocampus and frontal cortex of AD brains [89]. In transgenic APP23 mice who develop extensive A β pathology, the alpha 1-adrenoceptor antagonist prazosin reduced the generation of $A\beta$ in N2a cells and prevented memory deficits over time, potentially owing to the anti-inflammatory response induced by prazosin [90]. Prazosin antagonizes norepinephrine effects at brain postsynaptic alpha-1 adrenergic receptors, leading to the compensatory upregulation of existing locus coeruleus neurons that maintain age-appropriate increased levels of norepinephrine outflow in the central nervous system [90]. Due to the relevance of noradrenergic systems in AD and the potential effects on BPSD, Wang et al. investigated the tolerability and efficacy of prazosin for behavioral symptoms in patients with agitation/aggression in patients with AD. In a double-blind, parallel-group study, 22 patients were randomized to receive either 1-6 mg of prazosin daily or a placebo. After 8 weeks, participants who received prazosin demonstrated significant improvements in their agitated/ aggressive symptoms compared with those in the placebo group [91]. The onset of action with prazosin occurred rapidly, appeared to be clinically significant, and the degree of behavioral improvement matched changes observed in atypical antipsychotic randomized controlled trials [91]. Though symptoms were reduced and adverse effects were similar between the prazosin and placebo groups, the small sample size and high dropout rate during follow-up are constraints that limit interpretation of the study. Nonetheless, this study provides a rationale for further studies with larger samples, increased follow-up times, and analysis of efficacy for specific behavioral symptoms. There are currently no studies in progress for this purpose though prazosin is being used in a phase IV trial to increase the removal of tau and $A\beta$ from the brains of veterans who have experienced mild traumatic brain injuries and have an increased risk of developing AD (NCT03221751). Additionally, Wang and colleagues undertook a 12-week, randomized, double-blind, controlled trial for agitated patients with AD and suggested that prazosin use may aid in treating behavioral problems by reducing excess adrenaline in the brain (NCT01126099). No results from this latter study have been reported.

6.4 Lithium

Lithium salts are a mood stabilizer with anti-manic properties often used to treat bipolar disorder and other psychiatric disorders [92]. Despite the extensive use of lithium within the past 50 years, there remains ambiguity surrounding the general mechanisms producing the observed effects of the salts [93]. In the context of treating AD-related pathologies, lithium treatment potentially inhibits glycogen synthase kinase-3 β , which blocks the accumulation of A β peptides in the brains of transgenic mice otherwise programmed to experience an overproduction of amyloid precursor protein [94]. The inhibition of glycogen synthase kinase- 3β has been shown to reduce phosphorylated tau and inflammation in addition to increasing signaling activity of a serine/threonine kinase that regulates vital cellular functions such as growth and survival, protein synthesis, and transcription [94, 95]. Another potential mechanism postulates that lithium inhibits inositol monophosphatase, thereby reducing inositol triphosphate and altering the phosphatidylinositol signaling pathway needed for cell growth, death, secretion, and information processing [96]. This process upregulates autophagy and assists in the elimination of aggregate-prone cytosolic proteins [97]. Lithium treatment in animal models has been shown to improve or reverse aggression and learned helplessness behavior proposed to be related to depression, while reducing cognitive impairments [98-101]. Consequently, lithium may be a viable treatment for treating and preventing further AD-related neuropathologies. As lithium has been shown to significantly improve agitation, aggression, and restlessness in studies of psychiatric disorders, lithium may also benefit neuropsychiatric symptoms in AD. A phase II, 12-week, randomized, double-blind, controlled trial is currently in progress investigating the efficacy and side effects of low-dose lithium compared to a placebo in treating agitation in AD using changes in NPI scores as the primary outcome (NCT02129348). Additionally, another phase II, 12-week, double-blind, placebo-controlled trial is investigating the effects of a 600-mg/day dose of lithium in reducing agitation and aggression in patients with frontotemporal dementia as measured by the NPI (ClinicalTrials.gov identifier: NCT02862210).

6.5 Carbamazepine and Mirtazapine

Carbamazepine and mirtazapine are simultaneously being investigated for treatment of agitation and aggression associated with patients with AD in a large, phase III, multi-center study undergoing participant recruitment (Clinicaltrials.gov identifier: NCT03031184). Patients enrolled must be clinically diagnosed with AD according to the National Institute of Neurological and Communicative Disorders and Stroke and Alzheimer's Disease and Related Disorders Association criteria and have a CMAI score of 45 at baseline. Carbamazepine, whose mechanism of action is not fully elucidated, is an antiepileptic medication also prescribed for neuropathic pain and psychiatric disorders [102]. Mirtazapine is an antidepressant that acts by antagonizing noradrenergic and serotonergic receptors and is often prescribed to treat anxiety, insomnia, vomiting, and nausea [103]. Both drugs have previously demonstrated efficacious results for the indication being studied. A 12-week open-label pilot study found that a 15- to 30-mg daily dose of mirtazapine significantly reduced scores on the short-form CMAI [104]. Similarly, an open-label study found that carbamazepine was helpful in reducing agitated behavior in 15 patients with AD, who were not responsive to antipsychotics, based on measures of the Brief Psychiatric Rating Scale, although two patients discontinued because of adverse effects of leukopenia and allergic reactions [105]. Carbamazepine also improved agitation in patients with dementia as a secondary measure in a 6-week, randomized, single-blind, parallel-group study that enrolled 51 nursing home patients studying a daily dose of 300 mg [106]. The present study is examining the effects of 15 mg of mirtazapine per day and 100 mg of carbamazepine per day on changes in CMAI scores over 6 and 12 weeks.

6.6 Mp 101

MP-101 is currently in phase II clinical trials for the treatment of dementia-related psychosis and/or agitation, as defined as meeting clinical criteria for either AD, dementia with Lewy bodies, dementia associated with Parkinson's disease, frontotemporal lobe dementia, or vascular dementia, in addition to having a score of >4 in either the hallucinations or delusions items of the NPI, a score of ≥ 6 on the NPI Psychosis Subscale, or a score ≥ 6 in the NPI agitation/ aggression domain. The primary endpoint for this study is the change in NPI-Psychosis Subscale score from baseline (ClinicalTrials.gov identifier: NCT03044249). As a mitochondrial modulator, MP-101 represents a novel mechanism of action being studied for this indication. Preclinical studies conducted by the drug manufacturer reported that a daily dose of MP-101 restored mitochondrial function by reducing free radicals, increased brain-derived neurotrophic factor levels, and had protective effects on dopamine neurons in Parkinson's disease models [107]. The manufacturer also reported that MP-101 significantly reduced oxidative stress, preserved brain volume, and improved motor symptoms in preclinical models of Huntington's disease [108]. Given that mitochondrial deregulation has been implicated in neurodegenerative processes, MP-101 represents another potential treatment for reducing neuropsychiatric symptoms in dementia [108].

6.7 Bupropion/Dextromethorphan

Another agent currently being studied for treating agitation/ aggression in patients with AD is bupropion/dextromethorphan. A combination of bupropion and dextromethorphan, the mechanism of action and rationale is similar to dextromethorphan/quinidine and deudextromorphan [109]. The primary purpose is for dextromethorphan to exert its effects on *N*-methyl-D-aspartate receptors, σ 1 receptors, serotonin/ norepinephrine reuptake transporters, and neuronal nicotinic α 3 β 4 receptors in the central nervous system, mediated by bupropion inhibiting its metabolism [70, 109]. Bupropion is an antidepressant and is also indicated for smoking cessation [110, 111]. It blocks norepinephrine and dopamine reuptake, seeming to have synergistic effects in formulation with dextromethorphan [109]. The study is a phase II/III, double-blind, placebo-controlled trial with three separate treatment arms comparing the efficacy of bupropion/dextromethorphan, bupropion, and placebo over 5 weeks (Clinicaltrials.gov identifier: NCT03226522). The primary endpoint is change in the CMAI score from baseline at 5 weeks.

6.8 Oxytocin

An ongoing trial is currently recruiting participants diagnosed with frontotemporal dementia. This phase II multicenter study is exploring the tolerability and efficacy of an intranasal formulation of the neuropeptide oxytocin (Syntocinon[®]), on reducing apathy in these patients over 20 weeks (ClinicalTrials.gov identifier: NCT03260920). Oxytocin has previously been shown to have prosocial effects in humans and to improve measures of social cognition in patients with schizophrenia [112]. A double-blind placebo-controlled trial of intranasal oxytocin showed that one dose of 24 IU significantly improved NPI scores in patients with behavioral variant frontotemporal dementia 8 h and 1 week after administration [113]. Doses of 24, 48, and 72 IU of this oxytocin formulation have also been shown to be safe in another phase I study [114]. Thus, intranasal oxytocin appears to be tolerable and has exciting implications for improving apathetic symptoms by way of its demonstrated prosocial effects. The study is investigating the effects of a low, medium, and high dose of intranasal oxytocin and is expected to be completed in 2021.

7 Conclusions

The comparative efficacy of the new compounds being evaluated in this review remains unclear because of significant variance between trials regarding inclusion criteria, target symptoms, study designs, and selected outcomes. Of the investigative agents outlined in this review though, the behavioral disturbance most widely targeted in clinical trials thus far has been agitation/aggression in patients with AD. The agents explored that have demonstrated efficacy belong to a diverse range of drug classes, from atypical antipsychotics, SSRIs, cannabinoids, dextromethorphan combinatory formulations, and others with miscellaneous mechanisms of action. For this indication, the newer atypical antipsychotics investigated, brexpiprazole and lumateperone, have progressed quite far in the drug development pipeline. Both are currently being evaluated in phase III clinical trials. As the use of currently available atypical antipsychotics appears associated with increased risks of cerebrovascular events and mortality in elderly populations, these newer agents may represent safer options in this drug class for treating agitation/aggressive symptoms in patients with AD, but this must be demonstrated through larger, appropriately powered, randomized controlled trials before widespread use can be recommended [19, 20]. Selective serotonin reuptake inhibitors are considered safer than antipsychotics, which is why they may be preferentially prescribed to treat agitated/ aggressive symptoms in patients with AD [54]. Currently being studied in a large phase III trial, escitalopram is a promising advancement to this class of agents commonly prescribed for treating agitation. Targeting the endocannabinoid system, the synthetic cannabinoids dronabinol and nabilone have been efficacious in reducing behavioral symptoms associated with dementia on the CMAI and NPI in placebo-controlled crossover trials [66, 67, 69]. Of the agents explored, the results of the most recent nabilone trial have had promising results in reducing agitated symptoms. The dextromethorphan/quinidine agents dextromethorphan/ quinidine and deudextromethorphan are novel additions to the treatments being explored for agitation associated with dementia. Additionally, the dextromethorphan/bupropion combination agent is also being explored in a phase II/III trial. Prazosin was first shown to significantly reduce NPI scores in a small cohort of agitated patients with AD after 8 weeks; it remains a valuable candidate for the treatment of agitation and aggression associated with dementia to explore in larger studies [89]. Furthermore, a phase III trial of carbamazepine in conjunction with mirtazapine is the only ongoing trial comparing the efficacy of two agents on reducing agitated symptoms. Last, MP-101 is a novel addition as the studies launched for the treatment of agitation are addressing a mitochondrial basis for behavioral issues in AD for the first time [106].

With the advent of newer atypical antipsychotics for treating BPSD, few have been explored for treating psychotic symptoms in patients with dementia. Pimavanserin has demonstrated efficacious results for improving symptoms of psychosis in participants with AD after 6 weeks [41]. Furthermore, pimavanserin was found to selectively improve NPI-NH psychosis subscores in patients who had more severe symptoms at baseline [41]. This suggests that pimavanserin may represent a safer alternative to traditional antipsychotics arguably for patients who will most likely require such pharmacotherapy. The ongoing phase III trial exploring the effects of pimavanserin for up to 26 weeks is currently underway and may help further elucidate the therapeutic efficacy of this agent over a longer period of time. Other symptoms associated with dementia explored as a primary outcome in clinical trials were apathy, in which methylphenidate and intranasal oxytocin are currently being investigated in clinical trials. These agents have different neural bases: methylphenidate being used to improve motivation and oxytocin influencing prosocial behavior as a neuropeptide [111]. Oxytocin is being studied for frontotemporal dementia in which diminished prosocial behavior is characteristic of this condition, whereas methylphenidate is being studied for AD. ADMET-2 will further elucidate the therapeutic benefit of methylphenidate in a longer period of 6 months of treatments in a larger cohort of participants.

The diversity of the agents for BPSD summarized in this review indicates the potential for more agents for treating BPSD to be approved in the future. With the plethora of information becoming available, numerous considerations must be made to the application of these results in clinical practice. Treatment protocols must be personalized. First, contributing social and psychological factors must be given weight in the treatment plan selection process. If non-pharmacological treatments are unsuccessful or not feasible, then pharmacological agents can be considered using emerging biomarker evidence to determine what might work best for each patient. In this way, clinical trials must be receptive to the heterogeneity of neuropsychiatric and behavioral symptoms of dementia, incorporating opportunities for the development of biomarkers that can aid as predictors for treatment response.

Future studies should continue to use appropriate, validated, behavioral outcome measures when assessing the efficacy of the treatments being studied. Clinical trials investigating treatments for neuropsychiatric symptoms could also benefit from the incorporation of a psychosocial intervention; for instance, the S-CitAD study currently includes a structured psychosocial intervention for participants with clinically significant agitation. Participants who do not show a response to the intervention are then randomized to receive either escitalopram or placebo. Additionally, the ADMET-2 study includes a psychosocial intervention for both its methylphenidate and placebo arms. The inclusion of psychosocial interventions in trials mimics the management in neuropsychiatric settings; thus, relating the efficacy of the agent to the clinical setting in which they will ultimately occur.

Compliance with Ethical Standards

Funding No sources of funding were provided for the preparation of this review article.

Conflict of Interest Krista L. Lanctôt has received research grants from the National Institute of Aging for "Apathy in Alzheimer's Disease

Methylphenidate Trial II (ADMET II)" and "Escitalopram for Agitation in Alzheimer's Disease (S-CitAD)", Alzheimer Drug Discovery Fund, the Alzheimer Society of Canada (Grant 15–17), Alzheimer's Association, Canadian Institutes of Health Research, AbbVie, Lundbeck, Pfizer, Sanofi-Aventis, Janssen-Ortho Inc., Roche, and Wyeth; and honoraria from Abbvie and Lundbeck. Nathan Herrmann has received research grants from the Alzheimer Drug Discovery Fund, the Alzheimer Society of Canada, the National Institutes of Health, Canadian Institute of Health Research, Lundbeck, and Roche; and consultation fees from Lilly, Merck, and Astellas. Mehnaz Ahmed, Marlene Malik and Johannes Teselink declare they have no conflicts of interest relevant to the content of this review article.

References

- 1. WHO. Dementia: a public health priority. Dementia. 2015;1-4.
- Gonfrier S, Andrieu S, Renaud D, Vellas B, Robert P. Course of neuropsychiatric symptoms during a 4-year follow up in the REAL-FR cohort. J Nutr Health Aging. 2012;16(2):134–7.
- Antonsdottir IM, Smith J, Keltz M, Porsteinsson AP. Advancements in the treatment of agitation in Alzheimer's disease. Expert Opin Pharmacother. 2015;16(11):1649–56. https://doi. org/10.1517/14656566.2015.1059422.
- Devshi R, Shaw S, Elliott-King J, Hogervorst E, Hiremath A, Velayudhan L, et al. Prevalence of behavioral and psychological symptoms of dementia in individuals with learning disabilities. Diagnostics (Basel). 2015;5(4):564–76.
- Masopust J, Protopopová D, Vališ M, Pavelek Z, Klímová B. Treatment of behavioral and psychological symptoms of dementias with psychopharmaceuticals: a review. Neuropsychiatr Dis Treat. 2018;14:1211–20.
- Steinberg M, Shao H, Zandi P, Lyketsos CG, Kathleen A, Norton MC, et al. In dementia: the Cache County Study. Int J Geriatr Psychiatry. 2010;23(2):170–7.
- Brodaty H, Donkin M. Family caregivers of people with dementia. Dialogues Clin Neurosci. 2009;11(2):217–28.
- Vaingankar JA, Chong SA, Abdin E, Picco L, Jeyagurunathan A, Seow E, et al. Behavioral and psychological symptoms of dementia: prevalence, symptom groups and their correlates in community-based older adults with dementia in Singapore. Int Psychogeriatrics. 2017;29(8):1363–76.
- Murman DL, Colenda CC. The economic impact of neuropsychiatric symptoms in Alzheimer's disease: can drugs ease the burden? PharmacoEconomics. 2006;23:227–42.
- Herrmann N, Lanctôt KL, Sambrook R, Lesnikova N, Hébert R, McCracken P, et al. The contribution of neuropsychiatric symptoms to the cost of dementia care. Int J Geriatr Psychiatry. 2006;21(10):972–6.
- Kales HC, Gitlin LN, Lyketsos CG. Assessment and management of behavioral and psychological symptoms of dementia. BMJ. 2015;2(350):h369. https://doi.org/10.1136/bmj.h369.
- Blanco-Silvente L, Castells X, Saez M, Barcelo MA, Garreolmo J, Vilalta-Franch J, Capella D. Discontinuation, efficacy, and safety of cholinesterase inhibitors for Alzheimer's disease: a meta-analysis and meta-regression of 43 randomized clinical trials enrolling 16 106 patients. Int J Neuropsychopharmacol. 2017;20(7):519–28.
- Wang J, Yu JT, Wang HF, Meng XF, Wang C, Tan CC, Tan L. Pharmacological treatment of neuropsychiatric symptoms in Alzheimer's disease: a systematic review and meta-analysis. J Neurol Neurosurg Psychiatry. 2015;86(1):101–9.
- Moore A, Patterson C, Lee L, Vedel I, Bergman H. Fourth Canadian Consensus Conference on the Diagnosis and Treatment of Dementia. Can Fam Physician. 2014;60:433–8.

- Vigen CLP, Mack WJ, Keefe R, Sano M, Sultzer D, Stroup S, et al. Cognitive effects of atypical antipsychotic medications in patients with Alzheimer's disease: outcomes from CATIE-AD. Am J Psychiatry. 2011;168(8):831–9.
- Alexopoulos GS, Jeste D V, Chung H, Carpenter D, Ross R, Docherty JP. The expert consensus guideline series. Treatment of dementia and its behavioral disturbances. Postgr Med. 2005; Spec No:6-22.
- 17. Tampi RR, Tampi DJ, Balachandran S, Srinivasan S. Antipsychotic use in dementia: a systematic review of benefits and risks from meta-analyses. Ther Adv Chronic Dis. 2016;7(5):229–45.
- Tran E, Dishman B. Citalopram-induced QTc prolongation: a brief review of the data. Ment Health Clin. 2012;2(6):139–41.
- Porsteinsson AP, Drye LT, Pollock BG, Devanand DP, Frangakis C, Ismail Z, et al. Effect of citalopram on agitation in Alzheimer disease: The CitAD randomized clinical trial. JAMA - J Am Med Assoc. 2014;311(7):682–91.
- Ballard CG, Waite J, Birks J. Atypical antipsychotics for aggression and psychosis in Alzheimer's disease. Cochrane Database Syst Rev. 2006;(1).
- Campanelli CM, Fick DM, Semla T, Beizer J. Potentially inappropriate medication use in older adults: the American Geriatrics Society 2012 Beers Criteria. J Am Geriatr Soc. 2012;60(4):616–31.
- Lai CKY. The merits and problems of Neuropsychiatric Inventory as an assessment tool in people with dementia and other neurological disorders. Clin Interv Aging. 2014;9:1051–61.
- Radloff LS. A self-report depression scale for research in the general population. Appl Psychol Meas. 1977;1:385–401.
- Kørner A, Lauritzen L, Abelskov K, Gulmann NC, Brodersen A-M, Wedervang-Jensen T, et al. Rating scales for depression in the elderly: external and internal validity. J Clin Psychiatry. 2007;68(3):384–9.
- Helmes E, Csapo KG, Short JA. Standardization and validation of the multidimensional oberservation scale for elderly subjects (MOSES). J Gerontol. 1987;42(4):395–405.
- Cohen-Mansfield J, Marx MS, Rosenthal AS. A description of agitation in a nursing home. J Gerontol. 1989;44(3):77–84.
- Mega M, Carusi DA, Cummings JL, Rosenberg-Thompson S, Gornbein J, Gray K. The Neuropsychiatric Inventory: comprehensive assessment of psychopathology in dementia. Neurology. 2012;44(12):2308.
- Michael M, Cain M, Zubenko GS, Kollar M, Rosen J, Burgio L, et al. The Pittsburgh Agitation Scale: a user-friendly instrument for rating agitation in dementia patients. Am J Geriatr Psychiatry. 2009;2(1):52–9.
- Mohammad D, Ellis C, Rau A, Rosenberg PB, Mintzer J, Ruthirakuhan M, Herrmann N, Lanctôt KL. Psychometric properties of apathy scales in dementia: a systematic review. J Alzheimers Dis. 2018;66(3):1065–82.
- Steinberg M, Lyketsos CG. Atypical antipsychotic use in patients with dementia: managing safety concerns. Am J Psychiatry. 2012;169(9):900–6.
- 31. Das S, Barnwal P, Winston AB, Mondal S, Saha I. Brexpiprazole: so far so good. Ther Adv Psychopharmacol. 2016;6(1):39–54.
- Diefenderfer LA, Luppa C. Brexpiprazole: a review of a new treatment option for schizophrenia and major depressive disorder. Ment Health Clin. 2017;7(5):207–12.
- 33. Lundbeck H, Otsuka. Otsuka and Lundbeck announce improvement of agitation symptoms related to about brexpiprazole. Corporate Release. 2017;613:1–7. https://www.lundbeck.com/ upload/us/files/pdf/2017_Releases/Otsuka%20and%20Lundbeck %20announce%20results%20of%20brexpiprazole%20on%20 symptoms%20of%20agitation%20related%20to%20Alzheime r%E2%80%99s-type%20dementia.pdf. Accessed 17 Oct 2018.

- 34. Scale SP, Mcginley E. Intra-Cellular Therapies announces positive top-line results from the first phase 3 trial of ITI-007 in patients with schizophrenia and confirms the unique pharmacology of ITI-007 in a separate positron emission tomography study. Corporate Release. 2015;8–11. http://ir.intracellulartherapi es.com/news-releases/news-release-details/intra-cellular-thera pies-announces-positive-top-line-results. Accessed 20 Oct 2018.
- Intra-Cellular Therapies (ICT). Intra-Cellular Therapies presents data on lumateperone, ITI-214 and ITI-333. Corporate Release. 2017;1–2. http://ir.intracellulartherapies.com/news-releases/ news-release-details/intra-cellular-therapies-presents-data-lumat eperone-iti-214-and. Accessed 22 Oct 2018.
- Zhang L, Hendrick J. The presynaptic D2 partial agonist lumateperone acts as a postsynaptic D2 antagonist. Matters. 2018;1–4. https://sciencematters.io/articles/10.19185/matters.2017120000 06. Accessed 24 Mar 2019.
- Hellebø O, Howell J, Hunt M. Corporate presentation: behavioral disturbances in dementia, including Alzheimer's Dis. 2007. http://ir.intracellulartherapies.com/static-files/93b08960-f01c-4864-aa22-8cadb3539753. Accessed 23 Oct 2018.
- Vanover K, Dmitrienko A, Glass S, Kozauer S, Saillard J, Weingart M. Lumateperone (ITI-007) for the treatment of schizophrenia: placebo-controlled clinical trials and open-label safety switching study. Schizophr Bull. 2018;44:341.
- Snyder GL, Vanover KE, Zhu H, Miller DB, O'Callaghan JP, Tomesch J, et al. Functional profile of a novel modulator of serotonin, dopamine, and glutamate neurotransmission. Psychopharmacology. 2015;232(3):605–21.
- Stahl SM. Mechanism of action of pimavanserin in Parkinson's disease psychosis: targeting serotonin 5HT2A and 5HT2C receptors. CNS Spectr. 2016;21(4):271–5.
- 41. Ballard C, Banister C, Khan Z, Cummings J, Demos G, Coate B, et al. Evaluation of the safety, tolerability, and efficacy of pimavanserin versus placebo in patients with Alzheimer's disease psychosis: a phase 2, randomised, placebo-controlled, double-blind study. Lancet Neurol. 2018;17(3):211–2.
- Cruz MP. Pimavanserin (Nuplazid): a treatment for hallucinations and delusions associated with Parkinson's disease. P T. 2017;42(6):368–71.
- 43. Trillo L, Das D, Hsieh W, Medina B, Moghadam S, Lin B, et al. Ascending monoaminergic systems alterations in Alzheimer's disease: translating basic science into clinical care. Neurosci Biobehav Rev. 2013;37(8):1363–79. https://doi.org/10.1016/j. neubiorev.2013.05.008.
- Lanctôt KL, Herrmann N, Mazzotta P. Role of serotonin in the behavioral and psychological symptoms of dementia. J Neuropsychiatry Clin Neurosci. 2001;13(1):5–21. https://doi. org/10.1176/jnp.13.1.5.
- 45. Mitchell RA, Herrmann N, Lanctôt KL. The role of dopamine in symptoms and treatment of apathy in Alzheimer's disease. CNS Neurosci Ther. 2011;17(5):411–27.
- Herrmann N, Lanctôt KL, Eryavec G, Khan LR. Noradrenergic activity is associated with response to pindolol in aggressive Alzheimer's disease patients. J Psychopharmacol. 2004;18(2):215–20.
- Herrmann N, Lanctôt KL, Eryavec G, Van Reekum R, Khan LR. Growth hormone response to clonidine predicts aggression in Alzheimer's disease. Psychoneuroendocrinology. 2004;29(9):1192–7.
- Lanctôt KL, Herrmann N, Eryavec G, Van Reekum R, Reed K, Naranjo CA. Central serotonergic activity is related to the aggressive behaviors of Alzheimer's disease. Neuropsychopharmacology. 2002;27(4):646–54.
- Lanctôt KL, Herrmann N, Van Reekum R, Eryavec G, Naranjo CA. Gender, aggression and serotonergic function are associated

with response to sertraline for behavioral disturbances in Alzheimer's disease. Int J Geriatr Psychiatry. 2002;17(6):531–41.

- Wilkins JM, Forester BP. Update on SSRI treatment for neuropsychiatric symptoms of dementia. Curr Psychiatry Rep. 2016;18(2):1–7.
- Joubert AF, Sánchez C, Larsen F. Citalopram. Hum Psychopharmacol Clin Exp. 2000;15(6):439–45.
- 52. Pollock BG, Mulsant BH, Rosen J, Sweet RA, Mazumdar S, Bharucha A, et al. Comparison of citalopram, perphenazine, and placebo for the acute treatment of psychosis and behavioral disturbances in hospitalized, demented patients. Res Pract Alzheimers Dis. 2003;7:180–6.
- Pollock BG, Mulsant BH, Rosen J, Mazumdar S, Blakesley RE, Houck PR, et al. A double-blind comparison of citalopram and risperidone for the treatment of behavioral and psychotic symptoms associated with dementia. Am J Geriatr Psychiatry. 2007;15(11):942–52. https://doi.org/10.1097/JGP.0b013e3180 cc1ff5.
- Viscogliosi G, Chiriac IM, Ettorre E. Efficacy and safety of citalopram compared to atypical antipsychotics on agitation in nursing home residents with Alzheimer dementia. J Am Med Dir Assoc. 2017;18(9):799–802. https://doi.org/10.1016/j.jamda .2017.06.010.
- 55. Herrmann N. Lack of clinically useful response predictors for treating aggression and agitation in Alzheimer's disease with citalopram. Evid Based Ment Health. 2016;19(4):e24-24. https ://doi.org/10.1136/eb-2016-102474.
- 56. Ho T, Pollock BG, Mulsant BH, Schantz O, Devanand DP, Mintzer JE, et al. R- and S-citalopram concentrations have differential effects on neuropsychiatric scores in elders with dementia and agitation. Br J Clin Pharmacol. 2016;3:784–92.
- Wu E, Greenberg PE, Yang E, Yu A, Haim Erder M. Comparison of escitalopram versus citalopram for the treatment of major depressive disorder in a geriatric population. Curr Med Res Opin. 2008;24(9):2587–95.
- Barak Y, Plopski I, Tadger S, Paleacu D. Escitalopram versus risperidone for the treatment of behavioral and psychotic symptoms associated with Alzheimer's disease: a randomized double-blind pilot study. Int Psychogeriatr. 2011;23(9):1515–9.
- An H, Choi B, Park K, Kim D-H, Yang DW, Hong CH, et al. the Effect of escitalopram on mood and cognition in depressive Alzheimer's disease subjects. Alzheimers Dement. 2016;12(7):P825.
- Sherman C, Ruthirakuhan M, Vieira D, Lanctot KL, Herrmann N. Cannabinoids for the treatment of neuropsychiatric symptoms, pain and weight loss in dementia. Curr Opin Psychiatry. 2018;31(2):40–146.
- Liu CS, Chau SA, Ruthirakuhan M, Lanctot KL, Herrmann N. Cannabinoids for the treatment of agitation and aggression in Alzheimer's disease. CNS Drugs. 2015;29(8):615–23.
- 62. Woodward MR, Harper DG, Stolyar A, Forester BP, Ellison JM. Dronabinol for the treatment of agitation and aggressive behavior in acutely hospitalized severely demented patients with noncognitive behavioral symptoms. Am J Geriatr Psychiatry. 2014;22(4):415–9. https://doi.org/10.1016/j.jagp.2012.11.022.
- Borgelt LM, Franson KL, Nussbaum AM, Wang GS. The pharmacologic and clinical effects of medical cannabis. Pharmacotherapy. 2013;33(2):195–209.
- Tambaro S, Bortolato M. Cannabinoid-related agents in the treatment of anxiety disorders: current knowledge and future perspectives. Recent Pat CNS Drug Discov. 2012;7(1):25–40.
- van den Elsen GA, Ahmed AI, Verkes RJ, Kramers C, Feuth T, Rosenberg PB, et al. Tetrahydrocannabinol for neuropsychiatric symptoms in dementia: a randomized controlled trial. Neurology. 2015;84(23):2338–46.
- 66. Shelef A, Barak Y, Berger U, et al. Safety and efficacy of medical cannabis oil for behavioral and psychological symptoms of

dementia: an-open label, add-on, pilot study. J Alzheimers Dis. 2016;51(1):15–9.

- Volicer L, Stelly M, McLaughlin J, Morris J, Volicer BJ. Effects of Dronabinol on anorexia and disturbed behavior in patients with Alzheimer's disease. Int J Geriatr Psychiatry. 2002;12(9):913–9.
- Walther S, Schupbach B, Seifritz E, Homan P, Strik W, S. W, et al. Randomized, controlled crossover trial of dronabinol, 2.5 mg, for agitation in 2 patients with dementia. J Clin Psychopharmacol. 2011;31(2):256–8.
- Forester B, Romano C, Quayle W, Georgakas J, Rosenberg PB. Pilot trial of dronabinol adjunctive treatment of agitation in Alzheimer's disease (Ad) (Thc-Ad). Alzheimers Dement. 2017;13(7):P940.
- Lanctot K, Ruthirakuhan M, Gallagher D, Sherman C, Abraham EH, Verhoeff NPLG, Herrmann N, et al. Nabilone significantly improves agitation/aggression in patients with moderate-to-severe AD: preliminary results of a placebo-controlled, double-blind, cross-over trial. Alzheimer's & Dementia. 2018;14(7):1385–7.
- Nguyen L, Thomas KL, Lucke-Wold BP, Cavendish JZ, Crowe MS, Matsumoto RR. Dextromethorphan: an update on its utility for neurological and neuropsychiatric disorders. Pharmacol Ther. 2016;159:1–22. https://doi.org/10.1016/j.pharmthera .2016.01.016.
- Taylor CP, Traynelis SF, Siffert J, Pope LE, Matsumoto RR. Pharmacology of dextromethorphan: relevance to dextromethorphan/quinidine (Nuedexta[®]) clinical use. Pharmacol Ther. 2016;164:170–82. https://doi.org/10.1016/j.pharmthera .2016.04.010.
- 73. Capon DA, Bochner F, Kerry N, Mikus G, Danz C, Somogyi AA. The influence of CYP2D6 polymorphism and quinidine on the disposition and antitussive effect of dextromethorphan in humans. Clin Pharmacol Ther. 1996;60(3):295–307.
- Cummings JL, Lyketsos CG, Peskind ER, Porsteinsson AP, Mintzer JE, Scharre DW, et al. Effect of dextromethorphan-quinidine on agitation in patients with Alzheimer disease dementia: a randomized clinical trial. JAMA. 2015;314(12):1242–54.
- Tung R. The development of deuterium-containing drugs. Innov Pharm Technol. 2010;32:24–8.
- 76. Cummings J, Lyketsos CG, Tariot P, Peskind E, Nguyen U. Dextromethorphan/quinidine (AVP-923). Efficacy and safety for treatment of agitation in persons with Alzheimer's disease: results from a phase 2 study (NCT01584440). 2015 AAGP Annual Meeting Peer to Peer Behavioral Activa. Am J Geriatr Psychiatry. 2015;23(3):S164–5. https://doi.org/10.1016/j. jagp.2014.12.171.
- 77. Garay RP, Grossberg GT. AVP-786 for the treatment of agitation in dementia of the Alzheimer's type. Expert Opin Investig Drugs. 2017;26(1):121–32. https://doi.org/10.1080/13543 784.2017.1267726.
- Mclaurin J, Golomb R, Jurewicz A, Antel JP, Fraser PE. Inositol stereoisomers stabilize an oligomeric aggregate of Alzheimer amyloid B-Peptide and inhibit AB -induced Toxicity. J Biol Chem. 2000;275(24):18495–502.
- Fenili D, Brown M, Rappaport R, McLaurin JA. Properties of scyllo-inositol as a therapeutic treatment of AD-like pathology. J Mol Med. 2007;85(6):603–11.
- Salloway S, Sperling R, Keren R, Porsteinsson AP, Van Dyck CH, Tariot PN, Crans G. A phase 2 randomized trial of ELND005, scyllo-inositol, in mild to moderate Alzheimer disease. Neurology. 2011;77(13):1253–62.
- Abushakra S, Boada Rovira M, Porsteinsson AP, Liang E, Pastrak A, Vellas B. Study of ELND005 in agitation and aggression of Alzheimer's disease (harmony-AD): diagnostic criteria and patient characteristics. Neurodegener Dis. 2015;15(1):802.

604

- Sadasivan S, Pond BB, Pani AK, Qu C, Jiao Y, Smeyne RJ. Methylphenidate exposure induces dopamine neuron loss and activation of microglia in the basal ganglia of mice. PLoS One. 2012;7(3):e33693.
- Prommer E. Methylphenidate: established and expanding roles in symptom management. Am J Hosp Palliat Med. 2012;29(6):483–90.
- Padala PR, Burke WJ, Shostrom VK, Bhatia SC, Wengel SP, Potter JF, et al. Methylphenidate for apathy and functional status in dementia of the Alzheimer type. Am J Geriatr Psychiatry. 2010;18(4):371–4.
- Lensing SY, Sullivan DH, Ramirez D, Dennis RA, Padala PR, Bopp MM, et al. Methylphenidate for apathy in communitydwelling older veterans with mild Alzheimer's disease: a double-blind, randomized, placebo-controlled trial. Am J Psychiatry. 2017;175(2):159–68.
- Rosenberg PPB, Lanctot KL, Drye LT, Scherer RW, Herrmann N, Mintzer JE, et al. Safety and efficacy of methylphenidate for apathy in Alzheimer's disease: a randomized, placebo-controlled trial. J Clin Psychiatry. 2013;74(8):810–6.
- Ruthirakuhan M, Herrmann N, Abraham EH, Chan S, Lanctot KL. Pharmacological interventions for apathy in Alzheimer's disease. Cochrane Database Syst Rev. 2018;5:CD012197.
- Scherer RW, Drye L, Mintzer J, Lanctôt K, Rosenberg P, Herrmann N, et al. The Apathy in Dementia Methylphenidate Trial 2 (ADMET 2): study protocol for a randomized controlled trial. Trials. 2018;19(1):1–15.
- Van Der Mussele S, Mariën P, Saerens J, Somers N, Goeman J, De Deyn PP, et al. Behavioral syndromes in mild cognitive impairment and Alzheimer's disease. J Alzheimer's Dis. 2014;38(2):319–29.
- Katsouri L, Vizcaychipi MP, McArthur S, Harrison I, Suárez-Calvet M, Lleo A, et al. Prazosin, an αl-adrenoceptor antagonist, prevents memory deterioration in the APP23 transgenic mouse model of Alzheimer's disease. Neurobiol Aging. 2013;34(4):1105–15.
- 91. Wang LY, Shofer JB, Rohde K, Hart KL, Hoff DJ, McFall YH, Peskind ER. Prazosin for the treatment of behavioral symptoms in Alzheimer's disease patients with agitation and aggression. Am J Geriatr Psychiatry. 2010;17(9):744–51.
- Shorter E. The history of lithium therapy. Bipolar Disord. 2009;11(Suppl. 2):4–9.
- Rapoport SI, Basselin M, Kim Hyung-Wook SRJ. Bipolar disorder and mechanisms of action of mood stabilizers. Brain Res Rev. 2009;61(2):185–209.
- Phiel CJ, Wilson CA, Lee VM-Y, Klein PS. GSK-3alpha regulates production of Alzheimer's disease amyloid-beta peptides. Nature. 2003;423(6938):435–9.
- 95. Findling RL, Frazier JA, Kafantaris V, Kowatch R, McClellan J, Pavuluri M, et al. The Collaborative Lithium Trials (CoLT): specific aims, methods, and implementation. Child Adolesc Psychiatry Ment Health. 2008;2(4):1–13.
- Brown KM, Tracy DK. Lithium: the pharmacodynamic actions of the amazing ion. Ther Adv Psychopharmacol. 2013;3(3):163–76.
- 97. Motoi Y, Shimada K, Ishiguro K, Hattori N. Lithium and autophagy. ACS Chem Neurosci. 2014;5(6):434–42.
- Yu F, Zhang Y, Chuang D-M. Lithium reduces BACE1 overexpression, beta amyloid accumulation, and spatial learning deficits in mice with traumatic brain injury. J Neurotrauma. 2012;29(13):2342–51.

- 99. Zhang X, Heng X, Li T, Li L, Yang D, Zhang X, et al. Long-term treatment with lithium alleviates memory deficits and reduces amyloid-beta production in an aged Alzheimer's disease transgenic mouse model. J Alzheimers Dis. 2011;24(4):739–49.
- Kovacsics CE, Gould TD. Shock-induced aggression in mice is modified by lithium. Pharmacol Biochem Behav. 2010;94(3):380-6.
- Redrobe JP, Bourin M. The effect of lithium administration in animal models of depression: a short review. Fundam Clin Pharmacol. 1999;13(3):293–9.
- Ambrósio AF, Soares-da-Silva P, Carvalho CM, Carvalho AP, Bailly S, Pocidalo J, et al. Mechanisms of Action of Carbamazepine and Its. J Mol Biol. 2002;57(5):203–35.
- Al-Majed A, Bakheit AH, Alharbi RM, Abdel Aziz HA. Mirtazapine. Profiles Drug Subst Excipients Relat Methodol. 2018;43:209-54.
- Cakir S, Kulaksizoglu IB. The efficacy of mirtazapine in agitated patients with Alzheimer's disease: a 12-week open-label pilot study. Neuropsychiatr Dis Treat. 2008;4(5):963–6.
- Lemke MR. Effect of carbamazepine on agitation and emotional lability associated with severe dementia. Eur Psychiatry. 1995;10(5):259–62.
- 106. Tariot PN, Erb R, Podgorski CA, Cox C, Patel S, Jakimovich L, Irvine C. Efficacy and tolerability of carbamazepine for agitation and aggression in dementia. Am J Psychiatry. 1998;155(1):54–61.
- 107. Mitochon Pharmaceuticals. Mitochon pharmaceuticals raises \$1.6M for clinical stage development of MP 101; the first mitochondrial targeted agent to treat neurodegenerative diseases [press release] (2016 May 17). https://www.mitochonpharma. com/assets/pdf/2016-04-mitochon-pharmaceuticals-press-relea se.pdf. Accessed 20 Sept 2018.
- Onyango IG, Dennis J, Khan SM. Mitochondrial dysfunction in Alzheimer's disease and the rationale for bioenergetics based therapies. Aging Dis. 2016;7(2):201. https://doi.org/10.14336/ AD.2015.1007.
- 109. Axsome Therapeutics. Agitation in patients with Alzheimer's disease (AD). 2016. http://axsome.com/axs-05/agitation-in-patie nts-with-alzheimers-disease-ad/. Accessed 20 Sept 2018.
- Patel K, Allen S, Haque MN, Angelescu I, Baumeister D, Tracy DK. Bupropion: a systematic review and meta-analysis of effectiveness as an antidepressant. Ther Adv Psychopharmacol. 2016;6(2):99–144. https://doi.org/10.1177/2045125316629071.
- 111. Hughes J, Stead L, Cahill K, Lancaster T. Antidepressants for smoking cessation (Review). Cochrane Libr. 2014;1:1–145. https ://doi.org/10.1002/14651858.CD000031.pub4/abstract.
- 112. Pedersen CA, Gibson CM, Rau SW, Salimi K, Smedley KL, Casey RL, Leserman J, Fredrik Jarskog L, Penn DL. Intranasal oxytocin reduces psychotic symptoms and improves Theory of Mind and social perception in schizophrenia. Schizophr Res. 2011;132(1):50–3.
- 113. Jesso S, Morlog D, Ross S, Pell MD, Pasternak SH, Mitchell DGV, Kertesz A, Finger EC. The effects of oxytocin on social cognition and behaviour in frontotemporal dementia. Brain. 2011;134(9):2493–501.
- 114. Finger EC, MacKinley J, Blair M, Oliver LD, Jesso S, Tartaglia MC, Borrie M, Wells J, Dziobek I, Pasternak S, Mitchell DGV, Rankin K, Kertesz A, Boxer A. Oxytocin for frontotemporal dementia: a randomized dose-finding study of safety and tolerability. Neurol. 2015;84(2):174–81.